Author's response to reviews

Title: Giant Invasive Spinal Schwannoma in Children: Case Report and Review of the Literature

Authors:

Sudhakar Vadivelu (sxvadive@texaschildrens.org)
Prashant Prasad (gpprasad@texaschildrens.org)
Adekunle M Adesina (amadesin@texaschildrens.org)
Eugene Kim (eskim@texaschildrens.org)
Thomas G Luerssen (tgluerss@texaschildrens.org)
Andrew Jea (ahjea@texaschildrens.org)

Version: 2 Date: 16 April 2013

Author's response to reviews: see over
April 16, 2013

Dear Editor,

Thank you for considering our paper, “Giant Invasive Spinal Schwannoma in Children: Case Report and Review of the Literature” for publishing. Because of its rarity, there have been few published reports describing clinical features and surgical outcomes, especially in the pediatric patient population. To the best of our knowledge, our case represents the fourth child with giant invasive spinal schwannoma reported in the literature. We believe this can significantly advance our understanding of this rare medical entity. This is an original manuscript, and no part of it has been published elsewhere. We appreciate the time you are taking to consider our manuscript, and we look forward to hearing from you.

Sincerely,

Andrew Jea MD